

Author's response to reviews

Title: Cost-effectiveness of a health-social partnership transitional program for post-discharge medical patients

Authors:

Frances KY Wong (hsfwong@inet.polyu.edu.hk)

June Chau (jchau@hkucc.hku.hk)

Ching So (jcso@hkucc.hku.hk)

Stanley KF Tam (tamkfs@ha.org.hk)

McGhee Sarah (smmcghee@hkucc.hku.hk)

Version: 2 **Date:** 3 July 2012

Author's response to reviews:

As instructed, we have formatted the abstract according to the guidelines for authors and included an abstract in the manuscript file.